Author's response to reviews

Title: Cause-specific inequalities in mortality in Scotland: two decades of change. A population-based study

Authors:

Alastair H Leyland (a.leyland@msoc.mrc.gla.ac.uk)
Ruth Dundas (r.dundas@msoc.mrc.gla.ac.uk)
Philip McLoone (p.mcloone@yahoo.co.uk)
F Andrew Boddy (faboddy@btinternet.com)

Version: 3 Date: 19 July 2007

Author's response to reviews:

In response to your request the following formatting changes have been made.

Formatting changes requested
-------------------------------

Title - Please use sentence case in the title, both in the manuscript file and the submission system (i.e. remove all unnecessary capitals). Please also include 'A population-based study' within the main title in the manuscript.

Comment: done.

Affiliations - Please remember to include 'UK' in the affiliations. Please also provide a full affiliation for the fourth author if appropriate (i.e. the affiliation before retirement if this is when the research took place).

Comment: 'UK' included in affiliation. The research took place after the retirement of the fourth author, and his affiliation therefore remains 'retired'.

Keywords section - Please remove the keywords section from the manuscript.

Comment: done.

Please only include the title on the title page of the manuscript.

Comment: done.

Section headings - Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals). For example Abstract not ABSTRACT.

Comment: done.

References - In the manuscript text, please place the reference before the full stop.

Comment: done.
Competing interests - Please move this to before the authors' contributions section.

Comment: done.

Table title and legend - The table title should appear above the table. The legend (or key) should appear below the table.

Comment: the tables and figures now have titles at the top, legends at the bottom.

Yours sincerely

Alastair H Leyland